Author's response to reviews

Title: Obstetrician-Gynecologists' Knowledge of Sickle Cell Disease Screening and Management

Authors:

Ijeoma Azonobi (iazonobi@msm.edu)
Britta L Anderson (anderson.britta.l@gmail.com)
Vanessa Byams (ver0@cdc.gov)
Althea Grant (ang4@cdc.gov)
Jay Schulkin (jschulkin@acog.org)

Version: 4 Date: 22 August 2014

Author's response to reviews:

Dear Editor,

Thank you for your comments and suggestions. The requested changes we have made are:

1. Edited the typing errors for lines 48, 52, 73, 172.
2. Added 3 references to Discussion section, as per suggestion. Those references are Labbe et al., 2005; Wilkins-Haug et al., 1999; Sutton et al., 1999.
3. Added a footnote that briefly explained what ACOG is
4. Added a footnote that briefly explained what CARN is
5. Recognized that there is a paucity not only in the United States but internationally as well (as requested by Reviewer Anne Matthews for Line 64)

Thank you,
The Authors